Author’s response to reviews

Title: Lung virome in mechanically ventilated patients: a pilot study

Authors:

Raquel Nazareth (raquelnazareth@gmail.com)
Maria Chasqueira (mjchasqueira@nms.unl.pt)
Maria Rodrigues (lucia.rodrigues@nms.unl.pt)
Carolina Paulino (mcpaulino@chlo.min-saude.pt)
Catarina Conceição (acconceicao@chlo.min-saude.pt)
Lia Lêdo (lledo@chlo.min-saude.pt)
Úrsula Segura (ursula.marques@hbeatrizangelo.pt)
Madalena Santos (madalena.a.s@netcabo.pt)
António Messias (antonio.messias@hbeatrizangelo.pt)
Pedro Póvoa (pedro.povoa@nms.unl.pt)
Paulo Paixão (paulo.paixao@nms.unl.pt)

Version: 4 Date: 05 Feb 2020

Author’s response to reviews:

5th February 2020

Dear Dr Maire Clayton

BMC Pulmonary Medicine

We would like to thank you for the opportunity to resubmit the revised version of our manuscript “Lung virome in mechanical ventilated patients: a pilot study” (PULM-D-19-00374).

Please find below our answers to the issues now mentioned by the editor.

Sincerely yours,
Editor Comments:

Currently, the contributions of authors CP, CC, LL, US, AM does not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

Answer: clarified in the section “Authors’ contributions”